Intensive therapy for moderate established rheumatoid arthritis: the TITRATE research programme

David L Scott,¹ Fowzia Ibrahim,¹ Harry Hill,² Brian Tom,³ Louise Prothero,¹ Rhiannon R Baggott,¹ Ailsa Bosworth,⁴ James B Galloway,¹ Sofia Georgopoulou,¹ Naomi Martin,¹ Isabel Neatrour,¹ Elena Nikiphorou,¹ Jackie Sturt,⁵ Allan Wailoo,² Frances MK Williams,⁶ Ruth Williams¹ and Heidi Lempp^{1*}

¹Centre for Rheumatic Diseases, Department of Inflammation Biology, School of Immunology and Microbial Sciences, King's College London, London, UK ²ScHARR Health Economics and Decision Science, The University of Sheffield, Sheffield, UK

³MRC Biostatistics Unit, University of Cambridge, Cambridge, UK

⁴National Rheumatoid Arthritis Society, Maidenhead, UK

⁵Department of Adult Nursing, Florence Nightingale Faculty of Nursing, Midwifery & Palliative Care, King's College London, London, UK

⁶Twin Research and Genetic Epidemiology, School of Life Course Sciences,

King's College London, St Thomas' Hospital, London, UK

*Corresponding author heidi.lempp@kcl.ac.uk

Declared competing interests of authors: James B Galloway has received personal fees and non-financial support from AbbVie Inc. (North Chicago, IL, USA) and from Pfizer Inc. (New York, NY, USA); personal fees from Bristol Myers Squibb[™] (New York, NY, USA) and from Janssen Pharmaceuticals (Beerse, Belgium); and grants and personal fees from Celgene Corporation (Summit, NJ, USA) and from UCB Pharma (Brussels, Belgium). Elena Nikiphorou has received speaker honoraria and has participated in advisory boards for Pfizer Inc., AbbVie Inc., Sanofi SA (Paris, France), Gilead Sciences, Inc. (Foster City, CA, USA), Celltrion (Incheon, Republic of Korea) and Eli Lilly and Company (Indianapolis, IN, USA) outside the submitted work. Jackie Sturt has received personal fees from Eli Lilly and Company, and grants and personal fees from Spirit Healthcare Ltd (Leicester, UK).

Published August 2021 DOI: 10.3310/pgfar09080

Plain English summary

The TITRATE research programme Programme Grants for Applied Research 2021; Vol. 9: No. 8 DOI: 10.3310/pgfar09080

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

nflamed joints in rheumatoid arthritis are unpleasant and reduce quality of life. Many drugs reduce this inflammation. The ideal treatment goal is remission. When patients with active rheumatoid arthritis are treated intensively using gradually increasing doses of combinations of drugs they achieve more remissions. But many patients live with moderately active disease. There is uncertainty as to whether or not these patients benefit from intensive management. Our research focused on this problem.

Part of the research involved working with patients to identify how best to give intensive management. Monthly sessions with nurses in rheumatology clinics providing supportive care together with a handbook for patients were positively received by patients.

The central part of the research was a trial to investigate whether or not intensive management helped patients with moderately active rheumatoid arthritis. The trial compared intensive management using drug therapy and supportive non-drug approaches given by specialist nurses with standard care. Intensive management increased the number of patients in remission after 1 year. With intensive management, approximately one-third of patients were in remission, compared with about one-sixth of patients receiving standard care. Intensive management also reduced patients' fatigue levels. In the trial, intensive management did not increase harms.

The benefits of intensive management need to be set against its additional costs. These were relatively high. In addition, not all patients benefited. Patients who had high pretreatment physical disability or who were substantially overweight usually did not achieve remission.

The final part of the research considered broader perspectives of intensive management. Studying all trials of this approach showed that many types of intensive management increase remissions. Reviews of routine care over the last two decades showed that treatment intensities have gradually increased and more patients now achieve remissions.

Overall, our research shows that intensive management benefits patients with moderately active rheumatoid arthritis; however, not all patients benefited and the treatment was relatively expensive.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-0610-10066. The contractual start date was in October 2012. The final report began editorial review in July 2019 and was accepted for publication in February 2021. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, NETSCC, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PGfAR programme or the Department of Health and Social Care.

Copyright © 2021 Scott *et al.* This work was produced by Scott *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk